Author’s response to reviews

Title: Dysuria due to benign prostatic hyperplasia of the median lobe with ketamine-associated uropathy in a young male

Authors:

Zhangcheng Liao (xylzcheng@163.com)
Zhao Wang (xywangz07@163.com)
Zhongyuan Jin (13667356013@163.com)
Zhengyan Tang (xytzyan@163.com)

Version: 2 Date: 19 Sep 2019

Author’s response to reviews:

Response to Editor and Reviewer

Response to Editor:

We would like to sincerely express our great appreciation to you for your and the reviewers’ thoughtful and valuable comments on our manuscript. We have carefully made corrections according to your requirement. Thank you for taking the time to review our manuscript and we really do appreciate it.

1. Please combine your Discussion and Conclusion into one section called 'Discussion and Conclusions' as per our submission guidelines. Please remove the separate 'Discussion' and 'Conclusions' headings from within this section.

Response: We have combined your Discussion and Conclusion into one section called 'Discussion and Conclusions' in our manuscript.

2. Please include a statement in your Funding section describing the role of the funding body in the design of the study, the collection, analysis, and interpretation of data and in writing the manuscript.

Response: This study was supported by the National Natural Science Foundation of China (Award Number: 81570627). Our research was funded by this government fund, including research staff labor costs, patient transportation subsidies, and future publication costs.

3. Please provide your Author contributions section as a paragraph rather than bullet points.

Response: We have revised Author contributions as follows:

ZcL: Data collection, Manuscript writing. ZW: Data collection. ZyJ: pathological diagnosis. ZyT: the doctor responsible for the patient’s diagnosis and treatment. All authors have read and approved the manuscript, and ensure that this is the case.

4. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
Response: We have included this statement in the Authors' contributions as follows:

ZcL: Data collection, Manuscript writing. ZW: Data collection. ZyJ: pathological diagnosis. ZyT: the doctor responsible for the patient’s diagnosis and treatment. All authors have read and approved the manuscript, and ensure that this is the case.

5. Please amend your 'Consent for publication' statement to read as follows: 'Written informed consent for the publication of potentially identifying information was obtained from the patient.'
Response: We have amended 'Consent for publication' statement according to your requirement.

6. Please remove the cover letter and CARE checklist documents from the end of your manuscript file as these are no longer needed in the publication process.
Response: We have removed the cover letter and CARE checklist documents.

7. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: We have uploaded our manuscript, tables and figures according to your requirements.

Response to Reviewer 2: We would like to sincerely express our great appreciation to you for constructive comments on our manuscript which improved our manuscript substantially. Thank you for taking the time to review our manuscript. We would like to express our gratitude for your comments again.